

**EASL** HEPATOLOGY

# Risk of cirrhosis-related complications in patients with advanced fibrosis following hepatitis C virus eradication

Adriaan J. van der Meer<sup>1,\*</sup>, Jordan J. Feld<sup>2</sup>, Harald Hofer<sup>3</sup>, Piero L. Almasio<sup>4</sup>, Vincenza Calvaruso<sup>4</sup>, Conrado M. Fernández-Rodríguez<sup>5</sup>, Soo Aleman<sup>6,7</sup>, Nathalie Ganne-Carrié<sup>8</sup>, Roberta D'Ambrosio<sup>9</sup>, Stanislas Pol<sup>10</sup>, Maria Trapero-Marugan<sup>11</sup>, Raoel Maan<sup>1</sup>, Ricardo Moreno-Otero<sup>11</sup>, Vincent Mallet<sup>10</sup>, Rolf Hultcrantz<sup>6</sup>, Ola Weiland<sup>7</sup>, Karoline Rutter<sup>3</sup>, Vito Di Marco<sup>4</sup>, Sonia Alonso<sup>5</sup>, Savino Bruno<sup>12</sup>, Massimo Colombo<sup>9</sup>, Robert J. de Knegt<sup>1</sup>, Bart J. Veldt<sup>1</sup>, Bettina E. Hansen<sup>1</sup>, Harry L.A. Janssen<sup>1,2</sup>

<sup>1</sup>Department of Gastroenterology and Hepatology, Erasmus MC University Medical Center Rotterdam, Rotterdam, The Netherlands; <sup>2</sup>The Toronto Centre for Liver Disease, University Health Network, Toronto, Ontario, Canada; <sup>3</sup>Department of Internal Medicine III, Division of Gastroenterology and Hepatology, Medical University of Vienna, Vienna, Austria; <sup>4</sup>Gastrointestinal & Liver Unit, Dipartimento Biomedico di Medicina Interna e Specialistica, University of Palermo, Palermo, Italy; <sup>5</sup>Unit of Gastroenterology and Liver Diseases, University Hospital Fundación Alcorcón, Madrid, Spain; <sup>6</sup>Departments of Gastroenterology and Hepatology, Karolinska Institutet at Karolinska University Hospital, Stockholm, Sweden; <sup>7</sup>Karolinska University Hospital Huddinge, Karolinska Institutet, Stockholm, Sweden; <sup>8</sup>Unité d'Hépatologie, APHP Hôspital Jean Verdier, Université Paris 13, Inserm UMR 1162, France; <sup>9</sup>A.M. and A. Migliavacca Center for Liver Disease, First Division of Gastroenterology, Fondazione IRCCS Cà Granda Ospedale Maggiore Policlinico, Università degli Studi di Milano, Milan, Italy; <sup>10</sup>Unité d'Hépatologie, APHP Hôpital Cochin, Université Paris Descartes, Inserm U1016, Paris, France; <sup>11</sup>Gastroenterology-Hepatology Department, University Hospital La Princesa and Princesa Research Institute, Autonomous University of Madrid, Madrid, Spain; <sup>12</sup>Department of Internal Medicine, Humanitas University and IRCCS Istituto Clinico Humanitas, Rozzano (Milan), Italy

**Background & Aims:** The risk of hepatocellular carcinoma (HCC) is reduced but not eradicated among patients with hepatitis C virus (HCV)-induced advanced hepatic fibrosis who attained sustained viral response (SVR). We aimed to assess the risk of cirrhosis-related complications in this specific group of patients. **Methods:** Data from previously reported Western cohort studies including patients with chronic HCV infection and bridging fibrosis or cirrhosis who attained SVR were pooled for survival analyses on the individual patient level. The primary endpoint was HCC and the secondary endpoint was clinical disease progression, defined as liver failure, HCC or death.

**Results**: Included were 1000 patients with SVR. Median age was 52.7 (IQR 45.1–59.7) years, 676 (68%) were male and 842 (85%) had cirrhosis. Median follow-up was 5.7 (IQR 2.9–8.0) years. Fifty-one patients developed HCC and 101 had clinical disease progression. The cumulative 8-year HCC incidence was 1.8 (95% CI 0.0–4.3) among patients with bridging fibrosis and 8.7% (95% CI 6.0–11.4) among those with cirrhosis (p = 0.058). Within the cirrhosis group, the 8-year HCC incidence was 2.6% (95% CI 0.0–5.5) among patients <45 years, 9.7% (95% CI 5.8–13.6) among patients from 45–60 years, and 12.2% (95% CI 5.3–19.1) among patients >60 years of age at start of therapy (p = 0.006).

E-mail address: a.vandermeer@erasmusmc.nl (A.J. van der Meer).



Multivariable Cox analyses indicated that higher age, lower platelet count and diabetes mellitus were independently associated with development of HCC. After 8 years 4.2% (95% CI 0.1–8.3) of patients with bridging fibrosis and 15.8% (95% CI 12.3–19.3) of patients with cirrhosis experienced clinical disease progression (p = 0.007).

**Conclusions:** Patients with HCV-induced cirrhosis and SVR showed an annual risk of approximately 1% for HCC and 2% for clinical disease progression. Therefore, to prevent HCC surveillance, chronic HCV infection should preferably be treated before cirrhosis has developed.

**Lay summary**: Patients with cirrhosis who were able to eradicate their chronic HCV infection remain at substantial risk of primary liver cancer. The risk of liver cancer increases with higher age, laboratory makers suggesting more severe liver disease, and presence of diabetes mellitus. Also after successful antiviral therapy patients with HCV-induced cirrhosis should thus remain included in follow-up for early detection of liver cancer.

© 2016 European Association for the Study of the Liver. Published by Elsevier B.V. All rights reserved.

### Introduction

Chronic infection with the hepatitis C virus (HCV) may lead to the development of hepatic fibrosis, which can ultimately progress to cirrhosis. Patients with cirrhosis have an increased risk to develop end-stage liver disease and hepatocellular carcinoma (HCC) [1]. A recent meta-analysis found that chronic HCV-infected patients with advanced hepatic fibrosis had an overall annual risk of 2.9% to experience liver failure, 3.2% to develop HCC and 2.7% to die

Keywords: Chronic hepatitis C; Cirrhosis; Sustained virological response; Hepatocellular carcinoma; Survival analysis; Fibrosis; Liver failure.

Received 24 December 2015; received in revised form 24 September 2016; accepted 13 October 2016; available online 22 October 2016

<sup>\*</sup> Corresponding author. Address: Erasmus MC, University Medical Center, Department of Gastroenterology and Hepatology, 's Gravendijkwal 230, Room Ha 206, 3015 CE Rotterdam, The Netherlands. Tel.: +31 10 703 5942; fax: +31 10 436 5916.

of liver-related causes [2]. According to the current guidelines these patients should be included in HCC surveillance programs [3,4]. In the forthcoming years, there will be an increase in the incidence of HCV-induced cirrhosis and its clinical complications in the Western world [5,6]. In fact, in the USA it is estimated that by 2030 almost half of the population with chronic HCV infection will have cirrhosis [5].

Antiviral therapy has been available since the early 1990s and is usually considered successful when a sustained virological response (SVR), defined as HCV RNA negativity 12 or 24 weeks after cessation of treatment, is attained. However, SVR rates have been disappointing with the interferon (IFN)-based regimens, especially among patients with advanced liver disease [7]. Even for this important subgroup of patients, however, the efficacy of antiviral therapy has improved tremendously due to the development of direct-acting antiviral drugs. Recently, various IFN-free regimens have actually showed cure rates well over 90 percent among patients with chronic HCV infection and bridging fibrosis or cirrhosis. Another advantage of these new treatment regimens concerns the good safety-profile, which should lead to an increase in treatment uptake. In the end, treating more patients is essential if we really want to reduce HCV-related morbidity and mortality [8]. Institution of HCV screening programs, such as birth-cohort screening in the USA, should therefore be considered as a major accomplishment as well [9,10].

Both the antiviral treatment improvements and implementation of screening for HCV infection are likely to increase the population with HCV-induced cirrhosis and SVR. Several studies have shown that the incidence of liver failure and HCC is markedly lower among patients with cirrhosis and SVR as compared to those without SVR [11-15]. However, these studies also indicated that the risk of HCC was not entirely eradicated even when the HCV infection was cured. Case reports have even described patients who were diagnosed with HCC up to 13 years following successful antiviral therapy [16]. To date, risk factors of HCC following SVR are largely unknown. Despite respectable follow-up durations, prior Western cohort studies did not include sufficient patients with SVR and HCC for subgroup analyses. This can be attributed to the low rate of SVR with IFN-based regimens among patients with cirrhosis as well as the low risk of cirrhosis-related complications following HCV eradication.

Here we present the results of a study which aimed to assess the risk of HCC, as well as liver disease progression in general, among Western patients with chronic HCV infection and advanced hepatic fibrosis following achievement of SVR. We also studied the baseline factors which were associated with these outcomes in this specific population.

### Patients and methods

The primary investigators of previously reported European and Canadian-based cohort studies including patients with chronic HCV infection and advanced hepatic fibrosis were invited to participate in the current study based on individual patient data [11,12,14,15,17–24]. All patients with bridging fibrosis or cirrhosis were eligible for inclusion if they had attained SVR with an IFN-based antiviral treatment regimen and if there was available follow-up data. Patients coinfected with human immunodeficiency virus or hepatitis B virus were excluded.

The datasets of the original retrospective cohort studies based on chart review were sent to a single investigator (AJM), who monitored and entered the data into a uniform database system. In case of any doubt regarding the variables in the original dataset, the principal investigator of the participating cohort was contacted to answer any queries before incorporating the data into the central database. We collected all available information on patient demographics (sex, age, body mass index [BMI]), stage of liver disease (bridging fibrosis or cirrhosis), antiviral treatment (type of IFN used), presence of diabetes mellitus (DM), a history of severe alcohol use, virology (HCV genotype, anti-hepatitis B core antigen [antiHBc] status) and start-of-treatment laboratory parameters (platelet count, bilirubin, albumin, aspartate aminotransferase [AST], alanine aminotransferase [ALT], gamma-glutamyltransferase [gGT]). Cirrhosis was determined by either histopathological assessment of liver biopsy, transient elastography, or clinical judgement at the discretion of the treating physician based on a combination of biochemical parameters, clinical signs of portal hypertension and/or radiologic findings consistent with cirrhosis.

The study was conducted in accordance with the guidelines of the Declaration of Helsinki and the principles of Good Clinical Practice. All original study protocols had been reviewed and approved by the local ethics committees, and the results of all these studies have been published in the international literature. Informed consent had previously been obtained according to the local regulations of the participating centers in the original cohort studies.

#### Clinical events

The primary outcome of the study was development of HCC. Clinical disease progression was the secondary endpoint, which was defined as the occurrence of either HCC, liver failure, liver transplantation or mortality. In case multiple events occurred in a single subject, only the first event contributed to this combined clinical outcome measure.

In accordance with international guidelines, the diagnosis of HCC was based on histopathological confirmation or two coincident imaging techniques (computed tomography, magnetic resonance imaging or contrast-enhanced ultrasonography) showing a focal lesion larger than 2 cm with arterial-phase hyper enhancement or one imaging technique showing a focal lesion larger than 2 cm with arterial-phase hyper enhancement in the presence of an  $\alpha$ -fetoprotein level greater than 400 ng/ml [25]. Liver failure was defined as an episode of either ascites, bleeding varices, jaundice or hepatic encephalopathy. Death was classified as liver-related or not liver-related. In general, death is considered liver-related in case this is due to end-stage HCC or because of deteriorating liver function, but was at the discretion of the treating physician who collected the data.

#### Statistical analysis

Follow-up started 24 weeks after the end-of-treatment, which was defined as time 0, since patients with undetectable HCV RNA at this time point were classified as having attained SVR. Patients who were diagnosed with HCC or patients who were lost to follow-up prior to 24 weeks after the end-of-treatment were not included in the analyses. In the survival analyses regarding clinical disease progression, patients who were diagnosed with HCC or who experienced liver failure prior to 24 week after the end-of-treatment were excluded. Patients were censored at their last date of follow-up.

The Kaplan-Meier method was used to assess the cumulative incidence of clinical events during follow-up, and differences according to categorical variables were assessed by the log-rank test. Univariable and multivariable Cox regression analyses were performed to assess which baseline variables were associated with the occurrence of clinical events. The proportionality assumption was checked by assessing if the effect of the variables on the outcome was changing over the log-transformed follow-up time for both continuous and categorical variables. Our multivariable models were constructed by considering all variables with a p value <0.1 in univariable analysis in a stepwise backward selection approach. Hereafter the variables outside of the model were introduced oneby-one to assess for possible confounding. The decision to keep the AST/ALT ratio in the model for HCC was made afterwards because of the borderline statistical significance, the effect size, and the biological plausibility of the association. As the period of inclusion was rather long, the analyses were stratified for the median year of inclusion. Adjustment for the type of IFN was considered as the chance of SVR is higher with pegylated (Peg)IFN as opposed to IFN. Because the hazard ratios of the variables in the model for HCC did hardly change by inclusion of the type of IFN, while this did result in overparameterization due to the limited number of HCC events, we decided to refrain from adjusting for type of IFN in the analyses of HCC.

The primary analyses concerned complete-case analyses. As a result of differences in data collection among the participating cohort studies, entire cohorts could drop out of the multivariable analyses in case the included baseline variables were not available. As sensitivity analyses, multiple imputation to impute missing values was performed with the iterative random number-based Markov Chain Monte Carlo method assuming multivariate normality with non-informative prior to using the SAS Proc MI package, by which 10 complete datasets were constructed in order to check the stability of the final multivariable Cox regression models.

All statistical tests were two-sided, and a *p* value <0.05 was considered to be statistically significant. The significance level for interactions was set at *p* = 0.01 to correct for multiple testing. SPSS version 22.0.1 (SPSS Inc., Chicago, IL, USA) and SAS 9.3 PROC GENMOD (SAS institute, Cary, NC, USA) were used for all statistical analyses.

### Results

### Study population

In total, 1000 patients with advanced hepatic fibrosis who had attained SVR were analysed from previously reported HCV treatment cohorts with long-term follow-up [11,12,14,15,17-24] (Supplementary Table 1). The median age was 52.7 years (interquartile range [IQR] 45.1-59.7) and 676 (68%) patients were male. Cirrhosis was present in 842 (85%) patients, others were classified as having bridging fibrosis. The stage of liver disease was based on liver histology in 896 (90%) patients and on transient elastography, presence of esophageal varices, and/or clinical judgement in 83 (8%) patients. For 21 (2%) patients data regarding the diagnostic modality on which the stage of liver disease was based was not available. Table 1 describes all baseline variables. The majority of patients were treated with a PegIFNbased regimen (n = 701, 70%). In five (<1%) patients a protease inhibitor was added. The year that successful antiviral therapy was initiated ranged from 1987 to 2010 (median 2002 [IQR 1999-2005]). Patients were followed for a median duration of 5.7 years (IQR 2.9-8.0).

#### Hepatocellular carcinoma

During follow-up, 51 patients were diagnosed with HCC, which resulted in a HCC rate of 0.90 per 100 person-years (95% confidence interval [CI] 0.67-1.18) (Table 2). The time interval between SVR and the diagnosis of HCC ranged from 0.2 to 11.8 years (median 5.8 years, IQR 2.6-7.4). The median age at the time of HCC was 65 years (IQR 57–69). The overall cumulative 8-year HCC incidence was 7.6% (95% CI 5.2-10.0). The cumulative 8-year HCC incidence was markedly higher among patients with cirrhosis (8.7%, 95% CI 6.0–11.4) as compared to patients who were classified as having bridging fibrosis (1.8%, 95% CI 0.0-4.3) (Fig. 1). Among patients with cirrhosis, those below the age of 45 at the start of therapy showed a cumulative 8-year HCC rate of 2.6% (95% CI 0.0-5.5), while this was 12.2% (95% CI 5.3-19.1) among patients above 60 years of age (Fig. 2). Three of the 158 (2%) patients classified as having bridging fibrosis were diagnosed with HCC after 3.1, 3.9 and 9.8 years following achievement of SVR. All were around 60 years of age at the start of therapy. None had DM at baseline and their BMI ranged from 23.3 to 25.8. The first two patients had no history of alcohol abuse, but data on alcohol use was not available for the third patient. Their baseline platelets ranged from 127 to  $166 \times 10^9$ /L.

Table 3 shows the results of the Cox regression analyses regarding HCC occurrence among the total group of patients. Significantly associated with HCC in univariable analyses were higher age, lower platelet count, higher AST/ALT ratio, and prior unsuccessful antiviral therapy. Gender did not influence the risk of HCC (p = 0.426). In multivariable analyses, the platelet count

# JOURNAL OF HEPATOLOGY

Table 1. Baseline characteristics.\*

	Overall
	(n = 1000)
Age, years, median (IQR) <sup>#</sup>	52.7 (45.1-59.7)
<45 years	241/998 (24)
45-60 years	526/998 (53)
>60 years	231/998 (23)
Males	676/1000 (68)
BMI, kg/m², median (IQR) <sup>¥</sup>	25.7 (23.3-28.4)
BMI >28	211/752 (28.1)
Disease stage	
Bridging fibrosis	153/995 (15)
Cirrhosis	842/995 (85)
HCV genotype	
1	459/918 (50)
2	220/918 (24)
3	181/918 (20)
4	49/918 (5)
Other	9/918 (1)
Type of treatment	
IFN (± ribavirin)	294/997 (30)
PegIFN (± ribavirin)	701/997 (70)
consensus IFN (+ ribavirin)	2/997 (<1)
Laboratory markers of liver disease severity, median	
(IQR) <sup>£</sup>	
Platelet count, $\times 10^9/L$	150 (116-194)
Albumin, g/L	42 (39-44)
Bilirubin, mmol/L	13.7 (10.3-17.1)
AST/ALT ratio	0.72 (0.57-0.90)
gGT, IU/L	56 (34-98)
Treatment naïve, n/of total (%)	624/972 (64)
Year treatment started <sup>≠</sup>	2002 (1999-
	2005)
Diabetes mellitus, n/of total (%)	117/819 (14)
History of severe alcohol use, n/of total (%)	128/575 (22)
AntiHBc positive, n/of total (%)	198/643 (31)

ALT, Alanine aminotransferase; antiHBc, anti-hepatitis B core antigen; AST, aspartate aminotransferase; BMI, body mass index; gGT, gamma-glutamyltransferase; HCV, hepatitis C virus; IQR, interquartile range; IFN, interferon; PegIFN, pegylated interferon.

Data are presented as No./Total No. (%) unless otherwise noted.

<sup>#</sup>Age was available in 998 ( $\sim$ 100%) patients.

<sup>4</sup> BMI was available in 642 (64%) patients. One of the participating cohorts had collected data on BMI as a categorical variable (BMI  $\leq 28 \text{ kg/m}^2 \text{ or }>28 \text{ kg/m}^2$ ), so that more data was available when BMI was categorized according to the 28 kg/m<sup>2</sup> cut-off.

 $^{\rm E}$ Platelet count was available in 955 (96%) patients, albumin in 808 (81%) patients, bilirubin in 813 (81%) patients, AST/ALT ratio in 801 (80%) patients and gGT in 700 (70%) patients.

<sup>£</sup>The year treatment started was available in 991 (99%) patients.

remained an independent predictor of HCC. Although the HR of the AST/ALT ratio (HR 1.04, 95% CI 1.00–1.09, p = 0.084) remained similar to that in univariable analysis, the association between the AST/ALT ratio and HCC fell just outside of the pre-specified significance level in the multivariable Cox model. Adjusted for age and the laboratory markers of liver disease severity, the association between DM was statistically significant (HR 2.36, 95% CI 1.02–5.42, p = 0.044). When added to the final model, being treatment naïve was no longer significantly associated with HCC (HR 0.71, 95% CI 0.34–1.48, p = 0.363). The proportional hazard assumptions were met and there were no statistically significant interactions between the variables included in the final model. Following imputation of missing data, which was performed as a sensitivity analysis, the estimated HRs were similar (Table 3).

# Table 2. Event rate per 100 person-years.

	Events No.	Observation period, person-years	Rate/100 person-years (95% CI)		
Hepatocellular carcinoma	51	5671	0.90 (0.67-1.18)		
Liver failure	26	5664	0.46 (0.30-0.67)		
All-cause mortality	56	5750	0.97 (0.74-1.26)		
Clinical disease progression	101	5592	1.80 (1.47-2.20)		

CI; confidence interval.



**Fig. 1. Incidence of hepatocellular carcinoma following sustained virological response according to the presence of bridging fibrosis or cirrhosis.** Kaplan-Meier graph for the cumulative incidence of hepatocellular carcinoma (HCC) according to the presence of bridging fibrosis or cirrhosis. The *p* value is based on the log-rank test.

## Clinical disease progression

In nine patients, liver failure occurred either before the start of therapy (n = 3), during therapy (n = 1) or after therapy but prior to achievement of SVR (n = 5). After exclusion of these patients, 26 of the 991 remaining patients experienced liver failure during follow-up (Table 2). Ascites was most frequently observed (n = 13, 50.0%), followed by variceal bleeding (n = 7, 26.9%), hepatic encephalopathy (n = 2, 7.7%) and jaundice (n = 1, 3.8%). Three (11.5%) patients had either ascites and/or encephalopathy, as detailed data on the specific event was not available. None of the patients with bridging fibrosis experienced liver failure. Among patients with cirrhosis, the 8-year cumulative incidence of liver failure was 4.1% (95% CI 2.3-5.9) (Fig. 3A). In three (11.5%) of the patients with liver failure, the event occurred within 6 months of the diagnosis of HCC or thereafter. Three patients, all with HCC, underwent liver transplantation. During the follow-up 56 (5.6%) patients died, which was registered to be due to liver-related causes in 24 (42.8%) patients and due to other causes in 25 (44.6%) patients (Table 2). For 7 (12.5%) patients the cause of death was not known. The cumulative 8year all-cause mortality rate was 3.3% (95% CI 0.0-7.0) among patients with bridging fibrosis vs. 9.6% (95% CI 6.7-12.5) among patients with cirrhosis at baseline (Fig. 3B).

In total, 101 (10.2%) patients showed clinical progression of their liver disease, resulting in a clinical disease progression rate of 1.80 per 100 person-years (95% CI 1.47–2.20) (Table 2). The first event was HCC in 47 (46.5%) patients, liver failure in 24



**Fig. 2. Incidence of hepatocellular carcinoma following sustained virological response according to age among patients with cirrhosis.** The incidence curves for hepatocellular carcinoma (HCC) were constructed using the Kaplan-Meier method. Twenty-four weeks after cessation of antiviral therapy was considered as time 0. Statistical significance was assessed with the log-rank test.

(23.8%) patients and death in 30 (29.7%) patients. The overall cumulative 8-year clinical disease progression rate was 14.0% (95% CI 11.1–16.9), which differed significantly among patients with bridging fibrosis (4.2, 95% CI 0.1–8.3) as compared to patients with cirrhosis (15.8, 95% CI 12.3–19.3) (Fig. 3C). Again, patients with cirrhosis below the age of 45 at the start of successful antiviral therapy showed a statistically significantly lower incidence of events at 8 years of follow-up (7.9, 95% CI 2.4–13.4) as compared to those between 45 and 65 years of age (17.1, 95% CI 12.2–22.0) or those above the age of 65 (20.7, 95% CI 12.3–29.1).

Cox regression analysis indicated that higher age, cirrhosis, lower platelet counts, and lower albumin levels were independently and significantly associated with clinical disease progression (Table 4). The associations between the variables and clinical disease progression were proportional over time, and there were no statistically significant interactions among the variables included in the final model. The final multivariable Cox regression model showed similar results in the imputation analyses.

## Discussion

This is the first Western study among patients with chronic HCV infection and advanced liver disease to assess the risk of cirrhosis-related complications following achievement of SVR. We found that the annual risk of HCC among patients with cirrhosis who cleared their HCV infection with IFN-based antiviral therapy was almost 1%. The annual risk of cirrhosis-related morbidity or mortality, as a combined endpoint, was twice as high. The complications of advanced liver disease were more frequently observed among patients who were older at the start of successful antiviral therapy and among patients with more severe liver disease. As was previously described for patients with cirrhosis and ongoing HCV infection, DM was an independent risk factor for HCC following SVR as well [26].

Since the HCV-infected population is aging and the proportion of patients with more advanced liver disease is increasing, our

# JOURNAL OF HEPATOLOGY

Table 3. Cox proportional hazard analyses for hepatocellular carcinoma

	Hepatocellular carcinoma								
	Univariable analyses			Multivariable analyses (n = 630) <sup>#</sup>			Imputation analyses (n = 1000) <sup>#</sup>		
	HR	95% CI	p value	HR	95% CI	p value	HR	95% CI	p value
Age									
<45 years	1.00	Ref.	Ref.	1.00	Ref.	Ref.	1.00	Ref.	Ref.
45-60 years	5.13	1.57-16.79	0.007	8.54	1.13-64.64	0.038	9.68	1.28-72.95	0.028
>60 years	6.95	2.03-23.76	0.002	8.91	1.12-70.79	0.039	9.76	1.23-77.77	0.031
Males	1.28	0.69-2.38	0.426	-	-	-	-	-	-
BMI, per 1.0 kg/m <sup>2</sup>	1.03	0.96-1.13	0.314	-	-	-	-	-	-
Cirrhosis	2.94	0.91-9.42	0.071	-	-	-	-	-	-
Laboratory markers of liver disease severity									
Platelet count, per $10 imes 10^9/L$	0.93	0.88-0.98	0.005	0.94	0.87-1.00	0.048	0.93	0.87-0.99	0.029
Bilirubin, per mmol/L	1.01	0.99-1.04	0.233	-	-	-	-	-	-
Albumin, per g/L	0.97	0.91-1.04	0.428	-	-	-	-	-	-
AST/ALT ratio, per 0.1	1.04	1.00-1.08	0.046	1.04	1.00-1.09	0.084	1.04	1.00-1.09	0.068
gGT, per 10 IU/L	1.02	0.99-1.04	0.143	-	-	-	-	-	-
Treatment naïve	0.39	0.22-0.71	0.002	-	-	-	-	-	-
Diabetes mellitus	1.90	0.91-4.00	0.090	2.36	1.02-5.42	0.044	2.27	0.98-5.29	0.057
History of severe alcohol use	0.89	0.40-1.97	0.774	-	-	-	-	-	-
Anti-HBc positive	1.16	0.60 - 2.25	0.655	-	-	-	-	-	-

BMI, body mass index; HCV, hepatitis C virus; AST, aspartate aminotransferase; ALT, alanine aminotransferase; CI, confidence interval; gGT, gamma-glutamyltransferase; antiHBc, anti-hepatitis B core antigen.

\* The multivariable models were stratified by the median year of inclusion (2002).

results suggest that we will more frequently encounter HCC, liver failure and liver-related mortality among patients with cured HCV infection in the following years [5]. Although cirrhosis was not statistically significantly associated with HCC in our cohort, this is still likely to be an important risk factor. It is relevant to consider that not all participating cohorts included patients with bridging fibrosis, so that a lack of power may be of influence here. The platelet count, which represents the extent of liver damage and portal pressure, was significantly associated with HCC. Even though the AST/ALT ratio fell just outside of the predefined  $\alpha$ level in multivariable analyses, this readily available noninvasive marker of hepatic fibrosis showed a strong trend for a clinically relevant effect size. For this reason we kept the AST/ ALT ratio in our final model. The risk of clinical disease progression, of which more events could be registered, was statistically significantly higher in case of cirrhosis rather than bridging fibrosis. Regardless of the severity of the HCV-induced liver disease, the prevalence of insulin resistance is rapidly rising as well [27]. Despite accumulation of these unfavorable prognostic characteristics, improvements in antiviral therapy will enable us to cure more patients. In fact, various IFN-free regimens have been shown to cure approximately 95% of patients, whether or not cirrhosis is present [28–32]. Thus, it is highly relevant that physicians are aware of the continuing health risks following successful antiviral therapy among patients with HCV-induced advanced liver disease.

An important message of our study is that patients with HCV-related cirrhosis who have attained SVR should probably remain included in HCC surveillance programs. Current AASLD guidelines state that HCC surveillance is cost-effective in patients with hepatitis C if the annual risk exceeds 1.5% [3]. Although in our study the overall risk among patients with cirrhosis and SVR did not meet this cut-off, there are several reasons to justify why these patients should be followed. First, the risk of HCC increased with

age and the population with cured HCV infection is rapidly aging. In fact, patients who initiated their successful antiviral treatment course above the age of 60 did meet the annual HCC risk cut-off of 1.5% as the HCC rate in this subgroup was 1.50 per 100 personyears (data not shown). For patients who were younger at the time of SVR it is possible that as they age the risk of HCC may actually increase. Second, with the current effective and safe IFN-free treatment regimens patients who are older and patients with more advanced liver disease can be cured. Our data showed that more advanced hepatic fibrosis (lower platelet count, higher AST/ALT ratio and lower albumin level) were pre-treatment predictors of cirrhosis-related events following SVR. Also, given the intrinsic antiproliferative properties of IFN, we cannot exclude that the incidence of HCC will be higher among patients who attained SVR with IFN-free regimens. Third, in case of SVR, the potential gain in life-years following the diagnosis of HCC may be higher due to reduced risk of other cirrhosis-related complications [11-13,15,33]. In a randomized controlled study among patients with compensated HCV-induced cirrhosis and HCC, Shiratori et al. found that the 7-year survival following tumor ablation was 53% among those treated with IFN vs. 23% among those not treated with IFN [34]. Within the IFN-treated group, the survival at 7 years of follow-up was higher among patients with SVR (68%) as compared to patients without SVR (47%). Similar results were described in a retrospective study, which also analysed hepatitis C patients with cirrhosis who underwent IFN-based therapy prior to HCC development [35]. In this study the cumulative survival was 93% among those with SVR and 57% among those without SVR at 6 years following HCC. Further studies are required to determine the annual HCC risk cut-off for cost-effective surveillance among patients with cirrhosis and SVR.

In our study three patients with bridging fibrosis and SVR were diagnosed with HCC. Among these patients there was no



Fig. 3. Incidence of liver failure, all-cause mortality and clinical disease progression following sustained virological response according to the presence of bridging fibrosis or cirrhosis. The incidence curves for liver failure (A), all-cause mortality (B) and clinical disease progression (C) were constructed using the Kaplan-Meier method. Clinical disease progression was defined as the occurrence of liver failure, hepatocellular carcinoma, liver transplantation or death. Twenty-four weeks after cessation of antiviral therapy was considered as time 0. Statistical significance was assessed with the log-rank test.

clear evidence for other causes of liver injury. In one patient, however, the interval between the pre-treatment liver biopsy and the initiation of antiviral therapy was rather long (3.8 years) so that progression to cirrhosis cannot be excluded. In fact, the platelet counts were low in all three patients, which might suggest underestimation of severity of liver disease due to sampling error [36]. Nevertheless, cases of HCC have been previously reported in Western patients with chronic HCV infection without cirrhosis following SVR [37]. Although it cannot be excluded that successfully treated patients with bridging fibrosis may develop HCC, their risk seems to be low. Before deciding to discharge the individual patient with bridging fibrosis and SVR from further follow-up, physicians should take secondary markers of liver disease severity into account to ensure that the patient had not progressed to cirrhosis since the pre-treatment hepatic fibrosis assessment.

Although this is the first study to assess the incidence of HCC among a large cohort of patients with HCV-induced advanced fibrosis and SVR, previous studies have evaluated risk factors of HCC following SVR. Yet, all these studies were performed in East Asia and included predominantly patients without advanced liver disease [38-46]. Still, the reported cumulative HCC incidences were relatively high, with 5-year rates ranging from 1.5 to approximately 6%. Advanced hepatic fibrosis was an important universal risk factor for development of HCC following SVR in these studies. In this subgroup the cumulative HCC incidence ranged from 9 to 15.6% after 5 years of follow-up. Genetic differences, the earlier wave of HCV infection, the higher prevalence of hepatitis B virus and a higher dietary exposure to alcohol or aflatoxin through Aspergillus-contaminated food products in Asia are possible explanations for this epidemiological difference in HCC occurrence between the East and the West [47,48]. Our study is novel as it specifically assesses the residual risk of HCC, its pattern of development over time and specific risk factors following SVR in the subgroup of patients with cirrhosis from the West. We found age, severity of liver disease and diabetes to be associated with HCC occurrence among patients with advanced liver disease and SVR.

It will probably remain difficult to accurately identify patients with cirrhosis and SVR who have a sufficiently low risk of HCC to be discharged from further follow-up. Pre-treatment predictors may not suffice as there are many factors following SVR which can influence a patient's risk. Even though the current data might suggest that patients with cirrhosis below the age of 45 may not require surveillance following SVR, it remains unknown whether their risk of HCC will increase as they age. Future studies need to assess the predictive accuracy of post-treatment markers and their kinetics during follow-up, as a recent study indicated that regression of hepatic fibrosis following antiviral therapy among patients with cirrhosis was associated with improved clinical outcome [21]. A history of alcohol abuse was not associated with HCC or clinical disease progression in the current cohort. Although we cannot exclude the possibility that heterogeneity of the gathered data or missing data may be partly responsible for these lacking associations, it should also be considered that all included patients underwent IFN-based therapy which is not generally administered to those with severe alcohol abuse. Data on alcohol use following successful antiviral therapy were not available in our cohort. It may be expected, however, that continuous alcohol abuse increases the risk of cirrhosis-related complications among patients with advanced liver disease and SVR as well, so that also these patients should be advised to limit their alcohol intake [49,50]. Non-alcoholic fatty liver disease or new hepatotrophic infectious agents also represent plausible causes of continuing hepatic inflammation which may drive

# JOURNAL OF HEPATOLOGY

Table 4. Cox proportional hazard analyses for clinical disease progression.

#### **Clinical disease progression** Univariable analyses Multivariable analyses (n = 785)# Imputation analyses (n = 991)<sup>#</sup> HR 95% CI p value 95% CI 95% CI p value HR HR p value Age 1.00 Ref. Ref. 1.00 1.00 <45 years Ref. Ref. Ref. Ref. 45-60 years 2.68 1.41-5.12 0.003 1.81 0.87-3.76 0.112 2.08 1.08-4.01 0.028 >60 years 3.60 1.81-7.16 < 0.001 2.54 1.17-5.49 0.018 2.75 1.37-5.55 0.005 Males 1.21 0.79-1.86 0.390 BMI, per 1.0 kg/m<sup>2</sup> 1.02 0.96-1.09 0.465 2.94 1.29-6.73 0.010 2.99 1.05-8.51 0.040 2.78 1.17-6.63 Cirrhosis 0.021 Laboratory markers of liver disease severity Platelet count, per $10 \times 10^9/L$ 0.92 0.89-0.96 <0.001 0.020 0.95 0.040 0.95 0.91 - 0.990.91-1.00 Bilirubin, per mmol/L 1.02 1.00-1.03 0.012 Albumin, per g/L 0.93 0.90-0.97 0.002 0.94 0.90 - 0.990.018 0.94 0.90 - 0.990.016 AST/ALT ratio, per 0.1 1.04 1.01 - 1.060.006 ----gGT, per 10 IU/L 1.00-1.03 1.02 0.079 --Treatment naïve 0.71 0.47 - 1.070.050 Diabetes mellitus 0.94-2.72 0.085 1.60 History of severe alcohol use 0.80 0.44-1.45 0.464 . \_ 1.09 AntiHBc positive 0.68-1.77 0.715

BMI, body mass index; HCV, hepatitis C virus; AST, aspartate aminotransferase; ALT, alanine aminotransferase; CI, confidence interval; gGT, gamma-glutamyltransferase; antiHBc, anti-hepatitis B core antigen.

<sup>#</sup> The multivariable models were stratified by the median year of inclusion (2002) and adjusted for the type of interferon.

carcinogenesis following HCV eradication. Future studies need to assess their contribution to the outcome of patients with SVR in more detail.

Because the clinical efficacy of antiviral therapy is highest among those with advanced liver disease, costly new treatment regimens are mostly limited to this subgroup which is at highest risk of cirrhosis-related events [51]. The data presented here actually suggest that we should pursue successful antiviral therapy prior to the stage of advanced fibrosis. This can save the costs of both HCC surveillance as well as the cirrhosis-related complications. It might thus be worthwhile not to neglect patients with lesser stages of hepatic fibrosis, especially those prone to progress to cirrhosis [52].

Our study has several limitations. As data were combined from previously published cohorts from multiple countries, some heterogeneity regarding the collected data could not be prevented. On the other hand, the participating cohort studies had rather similar inclusion and exclusion criteria and comparable follow-up designs. The fact that many countries are represented in this combined cohort makes the presented estimates applicable to a broad spectrum of patients as well. Still, databases differed and there was missing data for some variables. We have therefore performed imputation analyses to check the stability of the estimated hazard ratios. Methodological differences between the participating cohorts with respect to data collection were an important reason for missing data in our study, however, as not all cohorts registered all the variables which were included in our analyses. As a result, possible bias due to missing data which might not be missing at random may be limited. Unfortunately, data regarding the evolution of liver disease during follow-up or at the time of HCC were not available, which might be relevant in order to differentiate patients with SVR who remain at risk for the cirrhosis-related complication from those who are no longer at risk. Also, because the cohort studies that were part of this analysis had retrospective designs, it is possible that some events may have been missed. This would tend to bias

our estimates to a lower rate of HCC, however, and thus only increase the need for ongoing surveillance. Finally, because all patients in this cohort were eligible to be treated with IFNbased therapy, the patients included here had advanced but clinically stable liver disease. While our data suggest that patients with more advanced cirrhosis prior to SVR are at higher risk of cirrhosis-related complications, future studies should assess the clinical outcome among those with Child-Pugh score B and C who are now successfully treated with the IFN-free regimens.

To conclude, patients with chronic HCV infection and advanced fibrosis who have attained SVR remain at risk of HCC and other cirrhosis-related complications. Preferably, patients should thus be cured of their HCV infection prior to the development of advanced hepatic fibrosis. In case this is not accomplished, it is important to consider continued HCC surveillance upon achievement of SVR.

### **Financial support**

The study was not sponsored by third parties in the sense of industrial partners or government grants. The study was funded by the Foundation for Liver and Gastrointestinal Research in Rotterdam. This foundation was established in 1985 as an institution without purpose of financial gain and has the mission to stimulate scientific research in the field of diseases of the liver and the gastrointestinal tract. A major part of its activities are undertaken at the Erasmus MC in Rotterdam, the Netherlands.

### **Conflict of interest**

Dr. Colombo reports personal fees from Gilead Sciences, personal fees from AbbVie, personal fees from BMS, personal fees from Roche, personal fees from Bayer, personal fees from Vertex, personal fees from Merck, personal fees from Novartis, personal fees

from Janssen, personal fees from Sanofi, outside the submitted work. Dr. Aleman reports personal fees from Consultant fees for lectures from Gilead, Abbvie and BMS, personal fees from Consulting from Gilead, Abbvie, BMS, MSD, outside the submitted work. Dr. D'Ambrosio reports personal fees from Gilead, personal fees from AbbVie, personal fees from Merck, outside the submitted work. Dr. Feld reports grants and personal fees from AbbVie, grants and personal fees from Gilead Sciences, personal fees from Bristol Meiers Squibb (BMS), grants and personal fees from Janssen, grants and personal fees from Merck, grants from Abbott, outside the submitted work. Dr. Ganne-Carrié reports personal fees from Gllead, personal fees from BMS, outside the submitted work. Dr. Hofer reports personal fees (lecture fees, advisory board) from AbbVie, BMS, Gilead, Janssen, MSD and a research grant form AbbVie outside the submitted work. Dr. Maan reports personal fees from AbbVie, during the conduct of the study. Dr. Mallet reports personal fees and non-financial support from Gilead, personal fees from Abbvie, personal fees and nonfinancial support from MSD, personal fees and non-financial support from JJ/Janssen-Cilag, personal fees from Bristol Myers Squibb, personal fees from Novartis, personal fees from Roche, outside the submitted work. Dr. van der Meer reports personal fees from Gilead, personal fees from BMS, outside the submitted work.

All other authors have no conflicts of interest with respect to presented work.

### Authors' contributions

All authors partook in study concept and design, acquisition of data and critical revision of the manuscript. Analysis and interpretation of data: A van der Meer, J Feld, B Hansen and H Janssen. Drafting of the manuscript: A van der Meer, J Feld, H Janssen. Statistical analyses: A van der Meer and B Hansen. Study supervision: A van der Meer and H Janssen.

### Supplementary data

Supplementary data associated with this article can be found, in the online version, at http://dx.doi.org/10.1016/j.jhep.2016.10. 017.

### References

- [1] Fattovich G, Giustina G, Degos F, et al. Morbidity and mortality in compensated cirrhosis type C: a retrospective follow-up study of 384 patients. Gastroenterology 1997;112:463–472.
- [2] Singal AG, Volk ML, Jensen D, et al. A sustained viral response is associated with reduced liver-related morbidity and mortality in patients with hepatitis C virus. Clin Gastroenterol Hepatol 2010;8:280–288.
- [3] Bruix J, Sherman M. Management of hepatocellular carcinoma: an update. Hepatology 2011;53:1020–1022.
- [4] European Association for the Study of the Liver. EASL Clinical Practice Guidelines: management of hepatitis C virus infection. J Hepatol 2011;55:245–264.
- [5] Davis GL, Alter MJ, El-Serag H, et al. Aging of hepatitis C virus (HCV)-infected persons in the United States: a multiple cohort model of HCV prevalence and disease progression. Gastroenterology 2010;138:513–521.
- [6] Razavi H, Waked I, Sarrazin C, et al. The present and future disease burden of hepatitis C virus (HCV) infection with today's treatment paradigm. J Viral Hepat 2014;21:34–59.

- [7] Vezali E, Aghemo A, Colombo M. A review of the treatment of chronic hepatitis C virus infection in cirrhosis. Clin Ther 2010;32:2117–2138.
- [8] Wedemeyer H, Duberg AS, Buti M, et al. Strategies to manage hepatitis C virus (HCV) disease burden. J Viral Hepat 2014;21:60–89.
- [9] Moyer VA. Force USPST. Screening for hepatitis C virus infection in adults: U. S. Preventive Services Task Force recommendation statement. Ann Intern Med 2013;159:349–357.
- [10] Smith BD, Morgan RL, Beckett GA, et al. Hepatitis C virus testing of persons born during 1945–1965: recommendations from the Centers for Disease Control and Prevention. Ann Intern Med 2012;157:817–822.
- [11] Aleman S, Rahbin N, Weiland O, et al. A risk for hepatocellular carcinoma persists long-term after sustained virologic response in patients with hepatitis C-associated liver cirrhosis. Clin Infect Dis 2013;57:230–236.
- [12] Bruno S, Stroffolini T, Colombo M, et al. Sustained virological response to interferon-alpha is associated with improved outcome in HCV-related cirrhosis: a retrospective study. Hepatology 2007;45:579–587.
- [13] Cardoso AC, Moucari R, Figueiredo-Mendes C, et al. Impact of peginterferon and ribavirin therapy on hepatocellular carcinoma: incidence and survival in hepatitis C patients with advanced fibrosis. J Hepatol 2010;52:652–657.
- [14] Fernandez-Rodriguez CM, Alonso S, Martinez SM, et al. Peginterferon plus ribavirin and sustained virological response in HCV-related cirrhosis: outcomes and factors predicting response. Am J Gastroenterol 2010;105:2164–2172.
- [15] Van der Meer AJ, Veldt BJ, Feld JJ, et al. Association between sustained virological response and all-cause mortality among patients with chronic hepatitis C and advanced hepatic fibrosis. JAMA 2012;308:2584–2593.
- [16] Mashitani T, Yoshiji H, Yamazaki M, et al. Development of hepatocellular carcinoma in a patient 13 years after sustained virological response to interferon against chronic hepatitis C: a case report. Cases J 2009;2:18.
- [17] Aghemo A, Rumi MG, Monico S, et al. The pattern of pegylated interferonalpha2b and ribavirin treatment failure in cirrhotic patients depends on hepatitis C virus genotype. Antivir Ther 2009;14:577–584.
- [18] Braks RE, Ganne-Carrie N, Fontaine H, et al. Effect of sustained virological response on long-term clinical outcome in 113 patients with compensated hepatitis C-related cirrhosis treated by interferon alpha and ribavirin. World J Gastroenterol 2007;13:5648–5653.
- [19] Calvaruso V, Bavetta MG, Ferraro D, et al. Risk of disease decompensation and HCC in patients with HCV cirrhosis non responders to PEG IFN plus RBV. Hepatology 2012;56:253A.
- [20] D'Ambrosio R, Aghemo A, Grazia Rumi M, et al. A morphometric and immunohistochemical study to assess the benefit of a sustained virological response in hepatitis C virus patients with cirrhosis. Hepatology 2012;56:532–543.
- [21] Mallet V, Gilgenkrantz H, Serpaggi J, et al. Brief communication: the relationship of regression of cirrhosis to outcome in chronic hepatitis C. Ann Intern Med 2008;149:399–403.
- [22] Rumi MG, Aghemo A, Prati GM, et al. Randomized study of peginterferonalpha2a plus ribavirin vs. peginterferon-alpha2b plus ribavirin in chronic hepatitis C. Gastroenterology 2010;138:108–115.
- [23] Rutter K, Stättermayer AF, Ferenci P, et al. Successful HCV eradication due to antiviral therapy is associated with improved long term outcome of patients with chronic hepatitis C. J Hepatol 2013;58:S369.
- [24] Trapero-Marugan M, Mendoza J, Chaparro M, et al. Long-term outcome of chronic hepatitis C patients with sustained virological response to peginterferon plus ribavirin. World J Gastroenterol 2011;17:493–498.
- [25] Bruix J, Sherman M. Management of hepatocellular carcinoma. Hepatology 2005;42:1208–1236.
- [26] Veldt BJ, Chen W, Heathcote EJ, et al. Increased risk of hepatocellular carcinoma among patients with hepatitis C cirrhosis and diabetes mellitus. Hepatology 2008;47:1856–1862.
- [27] Chen L, Magliano DJ, Zimmet PZ. The worldwide epidemiology of type 2 diabetes mellitus-present and future perspectives. Nat Rev Endocrinol 2012;8:228–236.
- [28] Afdhal N, Reddy KR, Nelson DR, et al. Ledipasvir and sofosbuvir for previously treated HCV genotype 1 infection. N Engl J Med 2014;370:1483–1493.
- [29] Afdhal N, Zeuzem S, Kwo P, et al. Ledipasvir and sofosbuvir for untreated HCV genotype 1 infection. N Engl J Med 2014;370:1889–1898.
- [30] Lawitz E, Gane E, Pearlman B, et al. Efficacy and safety of 12 weeks vs. 18 weeks of treatment with grazoprevir (MK-5172) and elbasvir (MK-8742) with or without ribavirin for hepatitis C virus genotype 1 infection in previously untreated patients with cirrhosis and patients with previous null response with or without cirrhosis (C-WORTHY): a randomised, open-label phase 2 trial. Lancet 2015;385:1075–1086.

# JOURNAL OF HEPATOLOGY

- [31] Lawitz E, Sulkowski MS, Ghalib R, et al. Simeprevir plus sofosbuvir, with or without ribavirin, to treat chronic infection with hepatitis C virus genotype 1 in non-responders to pegylated interferon and ribavirin and treatment-naive patients: the COSMOS randomised study. Lancet 2014;384:1756–1765.
- [32] Poordad F, Hezode C, Trinh R, et al. ABT-450/r-ombitasvir and dasabuvir with ribavirin for hepatitis C with cirrhosis. N Engl J Med 2014;370:1973–1982.
- [33] Morgan TR, Ghany MG, Kim HY, et al. Outcome of sustained virological responders with histologically advanced chronic hepatitis C. Hepatology 2010;52:833–844.
- [34] Shiratori Y, Shiina S, Teratani T, et al. Interferon therapy after tumor ablation improves prognosis in patients with hepatocellular carcinoma associated with hepatitis C virus. Ann Intern Med 2003;138:299–306.
- [35] Akamatsu M, Yoshida H, Shiina S, et al. Sustained viral response prolonged survival of patients with C-viral hepatocellular carcinoma. Liver Int 2006;26:536–542.
- [36] Rockey DC, Caldwell SH, Goodman ZD, et al. Liver biopsy. Hepatology 2009;49:1017–1044.
- [37] Sewell JL, Stick KM, Monto A. Hepatocellular carcinoma after sustained virologic response in hepatitis C patients without cirrhosis on a pretreatment liver biopsy. Eur J Gastroenterol Hepatol 2009;21:225–229.
- [38] Chang KC, Hung CH, Lu SN, et al. A novel predictive score for hepatocellular carcinoma development in patients with chronic hepatitis C after sustained response to pegylated interferon and ribavirin combination therapy. J Antimicrob Chemother 2012;67:2766–2772.
- [39] Hirakawa M, Ikeda K, Arase Y, et al. Hepatocarcinogenesis following HCV RNA eradication by interferon in chronic hepatitis patients. Intern Med 2008;47:1637–1643.
- [40] Ikeda M, Fujiyama S, Tanaka M, et al. Risk factors for development of hepatocellular carcinoma in patients with chronic hepatitis C after sustained response to interferon. J Gastroenterol 2005;40:148–156.
- [41] Iwasaki Y, Takaguchi K, Ikeda H, et al. Risk factors for hepatocellular carcinoma in Hepatitis C patients with sustained virologic response to interferon therapy. Liver Int 2004;24:603–610.

- [42] Makiyama A, Itoh Y, Kasahara A, et al. Characteristics of patients with chronic hepatitis C who develop hepatocellular carcinoma after a sustained response to interferon therapy. Cancer 2004;101:1616–1622.
- [43] Tokita H, Fukui H, Tanaka A, et al. Risk factors for the development of hepatocellular carcinoma among patients with chronic hepatitis C who achieved a sustained virological response to interferon therapy. J Gastroenterol Hepatol 2005;20:752–758.
- [44] Yamashita N, Ohho A, Yamasaki A, et al. Hepatocarcinogenesis in chronic hepatitis C patients achieving a sustained virological response to interferon: significance of lifelong periodic cancer screening for improving outcomes. J Gastroenterol 2014;49:1504–1513.
- [45] Arase Y, Kobayashi M, Suzuki F, et al. Effect of type 2 diabetes on risk for malignancies includes hepatocellular carcinoma in chronic hepatitis C. Hepatology 2013;57:964–973.
- [46] Asahina Y, Tsuchiya K, Nishimura T, et al. Alpha-fetoprotein levels after interferon therapy and risk of hepatocarcinogenesis in chronic hepatitis C. Hepatology 2013;58:1253–1262.
- [47] Ikeda K, Marusawa H, Osaki Y, et al. Antibody to hepatitis B core antigen and risk for hepatitis C-related hepatocellular carcinoma: a prospective study. Ann Intern Med 2007;146:649–656.
- [48] Liu Y, Wu F. Global burden of aflatoxin-induced hepatocellular carcinoma: a risk assessment. Environ Health Perspect 2010;118:818–824.
- [49] Pessione F, Ramond MJ, Peters L, et al. Five-year survival predictive factors in patients with excessive alcohol intake and cirrhosis. Effect of alcoholic hepatitis, smoking and abstinence. Liver Int 2003;23:45–53.
- [50] Verrill C, Markham H, Templeton A, et al. Alcohol-related cirrhosis-early abstinence is a key factor in prognosis, even in the most severe cases. Addiction 2009;104:768–774.
- [51] Van der Meer AJ, Veldt BJ, Feld JJ, et al. The number needed to treat to prevent mortality and cirrhosis-related complications among patients with cirrhosis and HCV genotype 1 infection. J Viral Hepat 2014;21:568–577.
- [52] Van der Meer AJ, Maan R, de Knegt RJ, et al. Natural history of HCV-induced liver disease. Curr Hepatitis Rep 2013;12:251–260.